Author’s response to reviews

Title: Prevalence of adjustment disorder among cancer patients, and the reach, effectiveness, costutility and budget impact of tailored psychological treatment: study protocol of a randomized controlled trial

Authors:

F.E. Beek (florie.van.beek@vu.nl)
L.M.A. Wijnhoven (Lonneke.Wijnhoven@radboudumc.nl)
F. Jansen (f.jansen1@amsterdamumc.nl)
J.A.E. Custers (Jose.Custers@radboudumc.nl)
E.J. Aukema (e.aukema@olvg.nl)
V.M.H. Coupé (V.Coupe@vumc.nl)
P. Cuijpers (p.cuijpers@vu.nl)
M.L. Lee (mvanderlee@hdi.nl)
B.I. Lissenberg-Witte (b.lissenberg@vumc.nl)
B. Wijnen (bwijnen@trimbos.nl)
J.B. Prins (judith.prins@radboudumc.nl)
Irma Verdonck-de Leeuw (im.verdonck@vumc.nl)

Version: 2 Date: 09 Dec 2019

Author’s response to reviews:

December 8th, 2019
Dear Maire Clayton,

Thank you for the careful handling of the manuscript entitled: “Prevalence of adjustment disorder among cancer patients, and the reach, effectiveness, cost-utility and budget impact of tailored psychological treatment: study protocol of a randomized controlled trial” (PSYO-D-19-00220R1). We were pleased to hear that the manuscripts is potentially acceptable for publication in BMC Psychology.

Please find enclosed the revised version of our manuscript and below our response to each point raised by the reviewers.

1. On the title page please list the full names for all authors.

We added the first name of each author on the title page.


2. In the Ethics approval and consent to participate please include whether permission has been obtained to access the Netherlands Cancer Registry.

We added the following sentence: “Permission has been obtained to access the Netherlands Cancer Registry.”

3. Please move the Ethics and dissemination section into the Ethics approval and consent to participate section.

This section is now moved to “Ethics approval and consent to participate”.

We have thoroughly re-examined the current version of the manuscript.

We hope that the revised version is accepted for publication in BMC psychology and look forward to hearing from you.
On behalf of all authors,

Sincerely,

Prof. dr. Irma M. Verdonck-de Leeuw

Amsterdam UMC, location VU University Medical Center
Dept of Otolaryngology-Head & Neck Surgery
P.O. Box 7057
1007 MB Amsterdam
Telephone: +31 20 4440931
E-mail: im.verdonck@vumc.nl